## Aalborg Universitet



### Digital product passports for a circular economy

Data needs for product life cycle decision-making

Jensen, Steffen Foldager; Kristensen, Jesper Hemdrup; Adamsen, Sofie; Christensen, Andreas; Wæhrens, Brian Vejrum

Published in: Sustainable Production and Consumption

DOI (link to publication from Publisher): 10.1016/j.spc.2023.02.021

Creative Commons License CC BY 4.0

Publication date: 2023

**Document Version** Publisher's PDF, also known as Version of record

Link to publication from Aalborg University

*Citation for published version (APA):* Jensen, S. F., Kristensen, J. H., Adamsen, S., Christensen, A., & Wæhrens, B. V. (2023). Digital product passports for a circular economy: Data needs for product life cycle decision-making. Sustainable Production and Consumption, 37, 242-255. https://doi.org/10.1016/j.spc.2023.02.021

#### **General rights**

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

- Users may download and print one copy of any publication from the public portal for the purpose of private study or research.
- You may not further distribute the material or use it for any profit-making activity or commercial gain You may freely distribute the URL identifying the publication in the public portal -

#### Take down policy

If you believe that this document breaches copyright please contact us at vbn@aub.aau.dk providing details, and we will remove access to the work immediately and investigate your claim.



Contents lists available at ScienceDirect

# Sustainable Production and Consumption



journal homepage: www.elsevier.com/locate/spc

# Digital product passports for a circular economy: Data needs for product life cycle decision-making

Steffen Foldager Jensen <sup>a,\*</sup>, Jesper Hemdrup Kristensen <sup>a</sup>, Sofie Adamsen <sup>a,b</sup>, Andreas Christensen <sup>a</sup>, Brian Vejrum Waehrens <sup>a</sup>

<sup>a</sup> Center for Industrial Production, Department of Materials and Production, Aalborg University, Pontoppidanstræde 103, 9220 Aalborg Øst, Denmark <sup>b</sup> Group Environment CoE, Group EHS, Grundfos A/S, Poul Due Jensens Vej 7, 8850 Bjerringbro, Denmark

#### ARTICLE INFO

Article history: Received 21 December 2022 Received in revised form 28 February 2023 Accepted 28 February 2023 Available online 4 March 2023

Editor: Prof. Ana Beatriz Lopes de Sousa Jabbour

Keywords: Closed-loop supply chain Sustainability Reverse logistics Digitalisation Environmental policy Policy instrument

#### ABSTRACT

Digital product passports have been proposed as a policy instrument to enable decision-making throughout product life cycles in favour of a circular economy. However, due to nascent conceptualisation and weak industrial embeddedness, the contents of such an instrument are a source of uncertainty. Situated in a mechatronics context, this multiple-case study explores the data needs for digital product passports. Extant research reveals seven data clusters: (1) usage and maintenance, (2) product identification, (3) products and materials, (4) guide-lines and manuals, (5) supply chain and reverse logistics, (6) environmental data and (7) compliance. To contextualise these clusters, interviews with three original equipment manufacturers (OEMs) as well as their respective customers, service partners, suppliers and third-party recycling companies were conducted. Through a survey, each specific data point was assessed in terms of importance, availability and sensitivity. The findings show differentiating needs for data across these actors, yet the exchange of data and its supporting infrastructure for closing resource loops remain at low maturity. Consequently, policymakers are recommended to roll out digital product passports in gradual stages, while industrial managers should proactively reconfigure data flows to account for decision-making in a reverse supply chain. Future research is encouraged to explore the use of digital product passports for decision-making in other industries.

© 2023 The Authors. Published by Elsevier Ltd on behalf of Institution of Chemical Engineers. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

#### 1. Introduction

Policy instruments are essential levers for sustainable production and consumption. As vehicles for promoting aspirational objectives, they directly or indirectly incentivise industries to comply with a shared trajectory towards sustainable development. In a European context, numerous instruments have been deployed, including extended producer responsibility (Subramanian et al., 2009), product environmental footprint (Pedersen and Remmen, 2022) and eco-design (MacDonald and She, 2015). Each instrument helps support a resource-effective and competitive economy in pursuit of climate neutrality by 2050, as outlined in the European Green Deal (European Commission, 2019). Under the Ecodesign for Sustainable Products Regulation, the most recent proposal concerns a digital product passport, which is expected to enable informed decision-making throughout product life cycles in support of a circular economy (European Commission, 2022). lenge. Although economically viable circular business concepts are starting to appear, particularly in the consumer electronics industry, many manufacturing companies remain hesitant to fully engage in a circular transformation due to plethoric (Ayati et al., 2022), entangled and context-dependent barriers (Jensen et al., 2022). Rather than being solely forward-oriented, supply chains must also be reconfigured to support a reverse flow of products, whereby products and materials are circulated at their highest utility, i.e. preserving as much embedded value as possible following 'the power of the inner circle' principle (Ellen MacArthur Foundation, 2012). Doing so calls for new forms of collaboration across supply-chain actors. Such collaborative arrangements build upon an alignment of interests, as well as a continuous exchange of data and information (Calicchio Berardi and Peregrino de Brito, 2021). However, a gap persists between the need for additional data and insufficient data sharing routines (Serna-Guerrero et al., 2022), arguably due to dispersed data and limited data sharing channels (Jäger-Roschko and Petersen, 2022) - despite increased attention in the role of a smart and digitally-enabled circular economy (Rosa et al., 2020). Consequently, actors find themselves lacking the requisite data

The proposal represents a reaction to a widespread industrial chal-

#### https://doi.org/10.1016/j.spc.2023.02.021

2352-5509/© 2023 The Authors. Published by Elsevier Ltd on behalf of Institution of Chemical Engineers. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

<sup>\*</sup> Corresponding author. E-mail address: steffenfj@mp.aau.dk (S.F. Jensen).

about product life cycles to effectively occupy a new role in a circular supply chain (Preston, 2012).

Digital product passports are expected to remedy this in the form of storage, allowing actors across the supply chain to insert and extract relevant product data and information. Despite increased attention, the industrial implications of a digital product passport are only just beginning to be explored; in the scientific discourse, the concept remains in its infancy. In the literature, focus is extensively directed towards the coherency and consistency of the policy instrument and its operational features to make it a valuable platform rather than an administrative burden (Götz et al., 2022). On that note, extant studies are concerned with proposing design options (Plociennik et al., 2022) and defining system requirements, including access control to protect intellectual property rights (Adisorn et al., 2021). To substantiate these arguments, further research on the contents of a digital product passport is needed (Adisorn et al., 2021; Berger et al., 2022; Götz et al., 2022). Data and information are likely to vary depending on the decision, use case and the actor who utilises it (Walden et al., 2021). Understanding these dynamics is expected to qualify subsequent discussions about system requirements and, ultimately, enable the industrial adoption of digital product passports. Consequently, this study addresses the following research question:

What are the critical decision points and data needs for actors in a circular supply chain to support product life cycle decision-making?

The remainder of the study is structured as follows. Section 2 outlines the critical decision points and data categories for a digital product passport, as evident in extant research. These two streams define a coding structure for the empirical data analysis. Section 3 describes the methodological and empirical foundation as well as the data collection and analysis methods. Section 4 presents the empirically identified data needs, which are then discussed further in Section 5. Section 6 provides concluding remarks.

#### 2. Conceptual background

This section delineates the conceptual background and academic discourse of the pillars upon which this research builds. First, we set the scene by introducing the notion of a smart circular economy. Second, we present the changing roles and decision-making contexts imposed on critical actors by the circular transition. Third, we introduce the notion of a digital product passport as a key enabler for operationalising circular strategies, including its contents, as presented in extant studies. In combination, these pillars construct a coding structure, and thus an analytical lens, through which the study is approached.

#### 2.1. Towards a smart circular economy

In the contemporary business landscape, many companies are chasing the potential for enhanced competitiveness from two transformative agendas: circular transformation and digital transformation (Ingemarsdotter et al., 2020). Despite having distinct value potentials, an increasing body of research connects the two agendas (Chauhan et al., 2022), often under the concept of a smart circular economy. Most dominantly, emerging digital technologies are capable of tracking and tracing products throughout their life cycle (Giovanardi et al., 2023), thus both enhancing supply-chain transparency and data generation in favour of a circular economy (Antikainen et al., 2018). Examples of this include the implementation of big data analytics for high-quality decision-making in pursuit of a circular economy (Awan et al., 2021; Bressanelli et al., 2018), the application of RFID tags to estimate the remaining lifetime of end-of-life products (Ondemir and Gupta, 2014) or the use of cyber-physical systems to cope with complex remanufacturing systems (Rejeb et al., 2022). While many of these consider unidirectional capabilities (i.e. digital technologies as enablers of a circular economy), others emphasise their synergetic effects, including the ability of a circular economy to enable digital transformation (Uhrenholt et al., 2022). Unlocking such synergetic effects enables companies to reconcile a forward-oriented and reverse supply chain through data-enabled decision-making (Pagoropoulos et al., 2017). Despite this increased attention, a research gap persists between the scientifically theorised potential and its industrial reality (Rosa et al., 2020), which prompts a characterisation of the smart circular economy as pre-paradigmatic (Kristoffersen et al., 2020; Pagoropoulos et al., 2017).

#### 2.2. Decision-making contexts in a circular economy

In contrast to a linear economy, which builds upon a 'take-make-dispose' system, a circular economy seeks to rethink production and consumption patterns through the lens of slowing, closing and narrowing resource loops at the micro, meso and macro levels in favour of social equity, environmental benefits and economic prosperity (Kirchherr et al., 2017). In the academic discourse, such ambitions are often substantialised through the utilisation of 'R-strategies', such as the 3Rstrategies (Reduce, Reuse, Recycle) (Ghisellini et al., 2016). Despite the variance in these frameworks, they are founded on a shared set of principles. First, companies should utilise their inner loops (Stahel, 2016) and strive to maintain functional value rather than recovering material value. Second, the multiple strategies are not mutually exclusive (Goyal et al., 2018); they are bound together by the principle of cascaded product and material flows. To circulate products at their highest utility, collaboration across the supply chain is considered a key enabler (Chen et al., 2017). Such activity is not necessarily limited to a dyadic relationship between two parties (e.g., a manufacturer and a supplier) but may unfold in a complex network of actors (Sudusinghe and Seuring, 2022), all of whom are assigned new roles since the supply chain is reversed and is no longer solely forward-oriented. Extant literature (e.g., Ellen MacArthur Foundation, 2012) highlights five groups of actors as being of particular relevance to the activities of forming a circular supply chain. Their roles and pertaining decision-making contexts are described in the following subsections and summarised in Table 1.

#### 2.2.1. Customers

Whether they are end-users or OEMs, customers can no longer uphold a passive role in terms of slowing and closing material loops (González-Sánchez et al., 2020), as they are expected to rethink a series of activities to increase circularity. First, procurement criteria must be expanded to not solely rely on traditional performance measures by integrating environmental properties and commitment and corporate social sustainability as decisive factors for more sustainable procurement (Tseng and Chiu, 2013), Second, along the use phase, customers are

#### Table 1

Decision-making contexts for a circular economy across supply-chain actors. The dotted line indicates that recycling companies differ from other actors as their change of focus does not take its offset from a linear vantage point.

	Supplier	Manufacturer	Service provider	Customer	Third-party recycling company
Linear supply chain	Forward-oriented production of components	Forward-oriented production of products	Time-based maintenance	Traditional procurement measures and disposal at end-of-life	Retaining material value with less regard for irreversible loss
Circular supply chain	Remanufacture of components	Selecting and employing value-retention strategies at product level	Condition-based maintenance	Sustainable procurement, product lifetime extension, end-of-life trajectories	Retaining material value with reversible loss

expected to extend product lifetimes and, eventually, explore end-oflife trajectories for products and materials to be circulated at their highest utility (Hopkinson et al., 2018).

#### 2.2.2. Service providers

From a circularity perspective, service providers have traditionally been responsible for prolonging product lifetimes through various maintenance and repair strategies, such as using spare parts to exchange defective components. Time-based maintenance has long been considered the most suitable strategy (Takata et al., 2004). However, with the purpose of providing more accurate service offers, conditionbased maintenance has gained traction, according to which decisions are based on data from monitoring product conditions (Ahmad and Kamaruddin, 2012). Although such a change of approach is not a new phenomenon and was hardly sparked by the circular transition, it draws upon the distinction between resource-effectiveness and resource-efficiency that is frequently highlighted in the circular economy discourse. For instance, Bockholt et al. (2020) argue that resource-effectiveness is tied to the utilisation of residual value (e.g., by recovering functional or material value), while resourceefficiency refers to the incremental process improvements of such activities.

#### 2.2.3. Manufacturers

Due to resource-intensive upstream activities, resource scarcity and increased focus on extended producer responsibility, manufacturing companies can no longer be solely concerned with a forward-oriented supply chain and logistics setup (Lüdeke-Freund et al., 2019). Instead, they must begin to decouple the consumption of virgin materials from business growth. In terms of closing resource loops, one essential measure relates to the development of product take-back systems, as well as capabilities to retain value at its highest utility (Geissdoerfer et al., 2017). Consequently, manufacturers must be able to select and employ suitable value-retention strategies at product level – assessing potential for products to be reused or remanufactured, or whether selected components or materials can be returned to suppliers or third-party recycling companies (Lopes de Sousa Jabbour et al., 2019).

#### 2.2.4. Suppliers

Rather than solely seeking material input from virgin resources, suppliers can improve sustainable value creation by developing capabilities to remanufacture post-market components (Xiong et al., 2016). The advantages of doing so include the utilisation of material expertise and existing facilities. In the early stages of circular transition, suppliers may find remanufacturing less attractive than manufacturers. However, as manufacturers increasingly explore remanufacturing, suppliers become incentivised to follow suit, as they would financially benefit from remanufacturing their post-market components over the alternative scenario, in which manufacturers would remanufacture the suppliers' components and thus reduce revenue for the supplier (Huang and Wang, 2017).

#### 2.2.5. Third-party recycling companies

Although some manufacturers and suppliers have facilities to recycle materials (e.g., to remelt aluminium), many have outsourced this activity to third-party recycling companies. As opposed to the other actors, recycling companies are born with a circular purpose, albeit one that is seen as a 'last resort' since only the material value is left to be retained (Bockholt et al., 2020). Nevertheless, recycling companies are also finding themselves in transition. Traditionally, the common understanding has been that 'all recycling is good recycling'. As such, these companies have been measured on the amounts of materials sent to recycling or the recovery rate (Graedel et al., 2011). Now, increasing attention is directed towards the utilisation and quality of recycling materials, and recycling companies are being encouraged to prepare materials for the highest utility on the market, thus conducting recycling with reversible loss of material properties whenever appropriate.

In sum, a variety of actors find themselves transitioning towards new roles. Although they are listed as having distinct intraorganisational responsibilities, significant value is to be found in interorganisational collaboration and increased exchange of data and information (Leising et al., 2018).

#### 2.3. Digital product passports: sharing routines and data needs

The most frequently mentioned construct in supply-chain collaboration for circular objectives refers to sharing data and information with collaborative partners (Sudusinghe and Seuring, 2022). Such routines are increasingly enabled by the proliferation of digital technologies, which are considered key levers of a circular transition (Uhrenholt et al., 2022) since the potential of collecting and utilising data is continuously expanding (Pagoropoulos et al., 2017). Consequently, an increasing yet scant body of literature aims to develop and demonstrate repositories for data to be utilised by actors across the supply chain for circulating products. Such repositories are increasingly conceptualised under the premise of 'passports' to accommodate the individual characteristics of their affiliation to products or materials. Being connected to each individual product, digital product passports are expected to serve as vessels for data sharing, as supply-chain actors - ranging from suppliers to third-party recycling companies - may both utilise and insert data to support each other in transitioning towards a circular supply chain (Adisorn et al., 2021). Across the industry reports and the academic discourse, however, the contents of such a digital product passport remain unclear and warrant further investigation (Adisorn et al., 2021; Götz et al., 2022). In general, the purpose of a product passport is taking shape, but it remains vaguely defined, particularly around which types of data and information are relevant to support decisionmaking across the value chain. Few reports and studies (Adisorn et al., 2021; Berger et al., 2022; Danish Business Authority, 2021; European Commission, 2022; Götz et al., 2022) point towards relevant data in their attempts to conceptualise digital product passports. Through their synthesis, seven data categories appear, as visualised in Table 2: (1) usage and maintenance, (2) product identification, (3) products and materials, (4) guidelines and manuals, (5) supply chain and reverse logistics, (6) environmental data and (7) compliance.

The high variety of data clearly reflects the multitude of factors that affect decision-making in a circular supply chain. Although these reports and studies serve to materialise the notion of a digital product passport, they share the limitation of having little industrial embeddedness and

#### Table 2

Data categories for digital product passports, as presented in academic studies and reports.

Data category	Usage and maintenance	Product identification	Products and materials	Guidelines and manuals	Supply chain and reverse logistics	Environmental data	Compliance
(Adisorn et al., 2021)		✓	✓	1			1
(Berger et al., 2022)	1	$\checkmark$	✓	1	1	$\checkmark$	1
(European Commission, 2022)	$\checkmark$		$\checkmark$			$\checkmark$	1
(Götz et al., 2022)		1	1	1	1	1	1
(Danish Business Authority, 2021)	1		1			1	

neglect multi-actor perspectives in terms of data needs. This leaves a gap, as these data needs are expected to differ across the supply chain. Further assessment of these studies reveals three elements of concern that underpin data needs and require further scrutinisation for the industrial adoption of digital product passports. First, perceived importance refers to the degree of criticality for decision-making (Berger et al., 2022). Second, availability of data has received attention, as companies already have vast amounts of data, but housed in dispersed data management systems (Jäger-Roschko and Petersen, 2022). Understanding the availability of data may affect the rollout of the policy instrument and elucidate the readiness of companies to implement digital product passports. Third, sensitivity of data is a cornerstone for operationalising digital product passports. While actors may have no objections to disclosing certain types of data to the general public, other types may be considered critical to the business and thus require digital product passports to contain multiple levels of access (Adisorn et al., 2021).

Synthesising this, the conceptualisation, and ultimately the implementation, of digital product passports is contingent on industrial insights into data needs. Despite the emerging scientific debate, how data needs differ among supply-chain actors remains understudied. Moreover, due to deficient industrial embeddedness in the debate, the nuances of the importance, availability and sensitivity of the respective data points remain absent. Together, these two gaps construct the scope of this study.

#### 3. Research approach

In response to the argument from Kristoffersen et al. (2020) that the area of data-enabled circular supply chains finds itself in a preparadigmatic stage, and that studies on digital product passports are scant with little industrial embeddedness, this study is explorative in nature and builds upon a multiple case-study methodology as described by Yin (2009). This aims to further strengthen the conditions for theorybuilding (Eisenhardt and Graebner, 2007), as it enables researchers to both examine the studied phenomena within and across settings (Baxter and Jack, 2008). Table 3 visualises the research design of the study, including its sources of data collection.

#### 3.1. Case selection

Three circular supply chains were selected as the empirical outset for the study, each of which consists of suppliers, the manufacturer itself as the focal company, service providers, customers and third-party recycling companies. Combined, they represent critical decisionmaking contexts throughout the life cycle of a specific product for which the digital product passport is intended. This selection rests on the rationale that if such a passport is to be utilised by several actors across a circular supply chain as a vessel for data and information sharing, then multi-actor perspectives are needed for the identification of its contents (i.e. the data and information needs).

The process towards this goal can be divided into two activities. The first step was to identify relevant manufacturing companies. For this, three selection criteria were defined: (1) the cases must all be actively engaged in circular transition and exploring the potential of end-of-life value retention; (2) the cases must produce products for which digital product passports are expected to become legally mandated; and (3) the cases should represent the same industry, as data needs and value retention opportunities are expected to differ across industries. Three manufacturing companies were identified, all of which are based in Denmark and are large producers of mechatronic products. Furthermore, from this point, manufacturers were utilised to contact

#### Table 3

Respondents for data collection. Asterisk (\*) refers to a response by a colleague, and superscripted letters (<sup>a,b,c</sup>) refer to joint interviews.

			Phase 1 – interviews		Phase 2 – survey
Role	Area	ID	Respondent	Duration	Response
Customer	Heat, ventilation, air conditioning, control technology	C1	Global product manager	35	
	Water treatment	C2	Circularity expert	40	Х
	Heat, ventilation, air conditioning, control technology	C3	Group sustainability manager	N/A	X*
		C3	Strategic sourcing specialist	25	Х
Service provider	Third-party service partner	SP1	Chief executive officer	30	
	Third-party service partner	SP2	After market manager <sup>a</sup>	25	Х
		SP2	Service technician <sup>a</sup>		Х
	Internal service unit	SP3	Senior service technician	30	Х
Manufacturer	Power conversion technology	M1	Project manager	30	Х
		M1	Project manager	35	Х
		M1	Engineering director	20	
		M1	Standardisation manager	35	
	Cooling and heating technology	M2	Post graduate	25	
		M2	Circular economy project director	30	
		M2	Senior project manager	25	Х
		M2	Head of Quality	30	Х
		M2	Complaint analyst	25	Х
		M2	Engineering director	30	
	Water pump solutions	M3	Global circular economy lead	35	Х
		M3	Sustainable manufacturing specialist <sup>b</sup>	40	Х
		M3	Lead engineer <sup>b</sup>		Х
		M3	Lead sustainability material specialist	35	Х
		M3	Project manager	30	Х
Supplier	Cast aluminium	S1	Senior EHS manager	20	Х
	Stainless steel	S2	Supplier sustainability manager <sup>c</sup>	50	Х
		S2	Senior environmental manager <sup>c</sup>		Х
	Cast iron	S3	Metallurgist	45	Х
Third-party recycling company	Metals, hazardous substances, plastics, electronics	R1	Waste specialist	25	Х
		R1	Key account manager	30	
	Electronics and plastics	R2	Sales & market director	40	Х
	Stainless steel	R3	Group sustainability officer	40	
			Total	14 h 20 m	N = 22 (71 %)

other actors. For example, key account managers were used to contact customers, while procurement managers were used to contact suppliers. Customers were selected based on their level of maturity in the circular transition, their attitude towards collaboration and their strategic importance. Suppliers were selected based on the criticality of components in terms of environmental footprint and economic value. Service providers and recycling companies were selected based on existing agreements and current collaborations with manufacturers. In sum, the study draws on two sampling techniques – convenience sampling and snowballing, as described by Taherdoost (2016) – to use present collaborative experiences as a vehicle for encouraging participation among external actors. Linkages between the actors are visualised in Table 4.

Notably, it is in relation to the manufacturers that the other actors enter a role as either supplier, customer, service provider or thirdparty recycling company. Acknowledging the interchangeable roles in a circular supply chain, where customers also become suppliers, and where recycling companies also become customers, it was decided for this study to refer to each actor according to its role in a forwardoriented supply chain for communicative purposes. Such empirical objectification enables coveted in-depth multi-actor perspectives on a scantly studied phenomenon that is contingent on several factors (e.g. industrial differences). Consequently, rather than pointing towards formal generalisability across industries, this study provides indicative results from a mechatronic industry to support further attempts to conceptualise digital product passports, thus aligning with Flyvbjerg (2006), who argues that such an approach has often paved the way for scientific progress and innovation.

#### 3.2. Data collection and analysis

The sample of actors enables an in-depth and multi-perspective exploration of data utilisation in a digital product passport. Studying such a complex socio-technical phenomenon calls for a qualitative approach. Consequently, the data for this study were collected through two phases.

**Phase 1:** The objective of this phase was to identify the data needs of a digital product passport. Semi-structured interviews were conducted with 28 practitioners, whose expertise was considered key to securing a circular supply chain. Interviews were conducted online. All interviews were recorded and subsequently transcribed, and interviewees as well as involved companies were all anonymised. In terms of data analysis, a conventional content analysis was selected as described by Hsieh and Shannon (2005), as this study seeks to explore a phenomenon with scant theoretical foundation. For this, the seven data clusters identified from the literature review (see Table 2) were used to create a coding structure. Within this structure, thematic codes emerged from scouting for patterns throughout the transcriptions, revealing the empirically derived data needs for a digital product passport.

**Phase 2:** Subsequent to the interviews, a survey was sent out to all interviewees. This served a twofold purpose. The survey helped validate and qualify the findings from the interviews by presenting initial results from the mapping of data needs. Furthermore, practitioners were asked to assess each data point in terms of relevance to them (as either user or provider) on a five-point Likert scale based on three parameters: (1) How *important* is a given data point to the actor utilising it, from 'not important' to 'of utmost importance'; (2) How *available* is a given data point to the actor providing it, from 'does not exist and would be difficult to generate' to 'data exist and are available'?; and (3) How

involved.

Table 4			
Linkages	between	the	actors



Fig. 1. Categories for assessing sensitivity.

sensitive is a given data point to the actor sharing it, from 'not sensitive at all' to 'highly sensitive'? Notably, rather than including all data points in every decision-making context, only those considered relevant to the respective decision are included. Consequently, the data points were passed through pre-sorting, as reflected in the high levels of importance. However, their relative importance differs, and coupling this to availability and sensitivity is expected to nuance the findings and enable a prioritisation of data points (e.g., by indicating critical requirements). Based on this assessment, the data points were then clustered into four groups:

*Out of scope* data points have low availability and low importance. These are not considered relevant for a digital product passport.

*Supportive* data points have high availability and low importance. As such data are readily available for sharing but of mediocre importance relative to others, they could be considered for a digital product passport but are not as essential to early phases of the rollout.

*Focused development* data have low availability and high importance. These data are particularly valuable for a given decision-making context, but they require extensive work to generate and structure the data by the industrial actors.

*Critical* data have high availability and high importance. Such types of data are considered critical contents of a digital product passport, as they are both available (or need minimal structuring) and add value to decision-making in favour of a circular supply chain.

Subsequently, sensitivity is added as an analytical layer to highlight the need to integrate multiple levels of accessibility into digital product passports, thus building confidence among industrial actors to engage in interorganisational data-exchange activities. A grey-scale colourcoding is applied to indicate this level of sensitivity, as illustrated in Fig. 1:

#### 4. Results

In this section, we present the empirical results. Here, it becomes evident that decision-making in favour of a circular economy builds upon two informative elements. First, several specific data points have been identified, including volume and recycled content. In other cases, a plurality of data points has been captured under a single category, such as labour conditions or environmental footprint. As an example, the latter covers a range of data points, such as carbon footprint, water consumption, land use change and ecotoxicity. However, such data points are combined into a single category to strike a balance between depth, breath and readability. Second, decision-making support is reliant on the availability of documents like product manuals, waste sorting guidelines and safety instructions. In the following subsections, the industrial needs for data and documents are connected to five use cases, as presented in Table 5. In doing so, we aim to concretise which types of data can support a circular supply chain while illustrating how they differ across various roles. Extending this mapping, we present results from assessing data needs in terms of their importance, availability and sensitivity.

	Supplier	Manufacturer	Service provider	Customer	Third-party recycling company
Supply chain 1	S1	M1	SP1	C1, C3	R1
Supply chain 2		M2	SP2	C2	R1
Supply chain 3	S2, S3	M3	SP3		R1, R2, R3

#### Table 5

Cross-referencing identified data points and documents with use cases. ( $\checkmark$ ) refers to the actor who is expected to utilise the data, while (\*) refers to the actor who is expected to provide the data.

	Customer: Sustainable procurement, product lifetime extension, end-of- life trajectories	Service provider: Condition-based maintenance	Manufacturer: Selecting and employing value- retention strategies at product level	Supplier: Remanufacture of components	Third-party recycling company: Retaining material value with reversible loss
Usage and maintenance					
Running hours and power (RH)		√	√*		
Location (LO)	*	$\checkmark$	√*		
External environment (EX)		√*	√*	✓	
Service log (SL)	*	√*	$\checkmark$		
Product identification					
Serial and product number (SN)		√	√*		
Manufacturing date (MD)		√	√*		
Country of origin (CO)	~	~	√*		
Products and materials	<u>г</u>				1
Bill of materials (BM)			<b>√</b> *		
Number of life cycles (NL)	✓		<b>√</b> *		
Volume (V)		,	<b>√</b> *	<i>.</i>	
Material composition (MC)		✓	<b>√</b> *	<b>√</b> *	✓
List of hazardous substances (LH)		√	√*	<b>√</b> *	~
Recycled content (RC)	~		<b>√</b> *	√*	
Compatibility of components across models (CC)		,	√*		
Average lifetime expectance (AL)	✓		√*		
Availability of spare parts (AS)	✓	✓	*		
Guidelines and manuals			(4)		
Guidelines for non-destructive disassembly (GD)	✓	✓	<b>√</b> *		~
Product manual (PM)	✓		√*		
Service manual (SM)	~	×	*		
Installation guideline (IG)	<b>√</b>	<b>√</b>	*		
Safety instructions (SI)	<b>√</b>	<b>v</b>	✓ * (*		
Waste sorting guidelines (WG)	~	V	√ *		~
Supply chain and reverse logistics			(*		1
Suppliers of components (SC)	~	/	V *		
Create mean rate market (CP)		v	V *		
Customer return channels (CR)	¥		<b>v</b>		
Environmental impacts				*	1
	*		•		I
Droduct specific regulation (RoHS/REACH) (DD)			*		
L abour conditions (LC)			*	*	

4.1. Customer: sustainable procurement, product lifetime extension and end-of-life trajectories

From a customer perspective, a digital product passport may serve a threefold purpose as a lever towards a circular supply chain. First, by increasing transparency across the supply chain, it is expected to support sustainable procurement in favour of all three dimensions of sustainability: social, environmental and financial. Doing so may support strategic targets for sustainable procurement and position customers more favourably for tenders, where sustainability is increasingly becoming a competitive parameter. To enable sustainable sourcing and procurement, a digital product passport is also expected to provide information about regulative compliance, for example, with regulation on labour conditions or with the RoHS Directive (Restriction of Hazardous Substances) and the REACH (Registration, Evaluation, Authorisation and Restriction of Chemicals) Directive. Additionally, customers deem it a suitable platform for suppliers to disclose environmental data. This includes life cycle impacts across various parameters, such as carbon footprint and water consumption, or related circularity aspects (such as percentages of recycled content or number of product life cycles), all of which are considered highly important. However, generating and structuring such data demands attention from upstream actors; having knowledge about this enables the customer to select the most sustainable alternative.

Second, customers are essential to prolonging product lifetimes through continuous maintenance and repair, for example, by utilising their own service functions. For an elaborate description of adherent data needs, see Section 4.2, as the two specific decision-making contexts are closely related.

The third scenario for which a digital product passport is considered valuable for customers revolves around end-of-life options. When a given product no longer serves its original purpose, it must be treated safely, and its materials must be circulated at their highest utility. Channels for this may vary. If the product manufacturer operates a take-back programme, a passport could serve as a means of informing customers about return channels, such as directing them towards consolidated pick-up stations or pre-paid return boxes. If not, materials are likely to be recycled. Often, third-party recycling companies reward companies that disassemble and sort materials into clean fractions. Being able to access relevant documents, including guidelines for non-destructive

disassembly as well as waste sorting instructions, would strengthen the ability of customers to handle electronic and mechanical waste responsibly.

The data needs to support all three of these decision-making contexts are combined in Fig. 2, which depicts the high demand for additional data from the customer perspective. Priority data include product compliance as well as guidelines, instructions and manuals, much of which are obtained already through procurement. While certain types of data related to slowing and closing resource loops exist (e.g., availability of spare parts or guidelines for non-destructive disassembly), a large share requires additional work from the product manufacturer, including in the fields of environmental footprint, customer return channels or tracking the number of life cycles.

#### 4.2. Service provider: condition-based maintenance

Service providers, as represented in this study, consider a digital product passport a coveted platform for increased data sharing to strengthen their service offers and repair activities. At the same time, they expect to occupy a dual role as both users and providers of data and information due to their linkage position between manufacturers and customers. At its very core, as service providers instigate a maintenance task, they stand to benefit from easily accessing product identification measures, including serial and product numbers as well as country of origin and date of manufacture, to log the activity. When assessing the task at hand, knowledge about the external environment in which the product has operated is an essential determinant when it comes to inspecting the product to detect faulty components as well as taking proper safety precautions. For instance, if a mechatronic device has been operating in a marine environment, saltwater is likely to have corroded its metallic components, increasing the risk of defective parts over time. Understanding such conditions enables the service partner to deliver the most optimal service, for example, by applying substances to prevent saltwater corrosion onwards in the product life cycle. Alternatively, from a safety concern, it is vital for a service provider of pump solutions, for example, to know whether a pump has distributed water or chemical substances. Currently, obtaining such knowledge comes from experience and specialisation. As service providers are often highly specialised, they will in time accumulate knowledge about specific environmental conditions, yet with little attention to structuring such a body of knowledge. Due to its value, not only for service providers themselves, but also for manufacturers (see Section 4.3), the digital product passport could support formalising knowledgesharing routines across supply-chain actors. Furthermore, as suggested by service providers, a passport would preferably provide data about the availability of spare parts and contain documents like a service manual and guidelines for non-destructive disassembly. In general, service partners see value in having all this information stored on a single platform. Therefore, they further suggest incorporating the service log into the digital product passport so that both they and the manufacturers can access the repair history of a product.

As shown in Fig. 3, priority data includes product identification measures and availability of spare parts, all of which hold low sensitivity. Furthermore, they cover a service manual and data related to materials (e.g., a list of hazardous substances and material composition), which hold higher degrees of sensitivity. Focused development data are characterised by more dynamic data, such as the service log, location and external environment, while installation guidelines are considered supportive data.

# 4.3. Manufacturer: selecting and employing value-retention strategies at product level

Manufacturers consider a digital product passport a promising vehicle for informing customers about potential channels for returning worn-out or defective products. This could involve either pointing to pre-paid envelopes or raising awareness about consolidated pick-up stations. When manufacturers have acquired products, they request extensive data about the use-phase of a product, as shown in Fig. 4. Practitioners deem digital product passports suitable for facilitating such data exchange. Key data and information include: When and where was the product installed, and how many hours has it been running? Has it operated at its highest capacity continuously or at different levels? Having knowledge about the running hours could – in combination with the estimated average product lifetime – provide an indicator of the remaining lifetime of a product. Moreover, gaining insight into



Fig. 2. Data to support customers in sustainable procurement, product lifetime extension and end-of-life trajectories.



Fig. 3. Data to support service providers in conducting condition-based maintenance.

the external environmental conditions under which the product has been operating is considered essential to its ability to undergo multiple life cycles. Examples of harmful environmental factors include the amount of dust to which the product has been exposed, levels of vibration and contact with potential hazardous substances, which may have contaminated materials. Such use-phase data are considered core constituents of the assessment as to whether returned products can be reused or remanufactured or sent to a third-party recycling company. As returned products are often either worn out or have failed on the market, direct reuse may not be preferable. Instead, manufacturers are inclined towards replacing worn-out or defective components, thus engaging in remanufacturing and refurbishing activities. Digital product passports are expected to enable this in multiple ways. Being able to access the service log, as updated by service providers or customers, to gain knowledge about the repair history is considered highly important to assess the lifetime of components, yet the availability of such data is rather



Fig. 4. Data to support manufacturers in selecting and employing value-retention strategies at product level.

low. As manufacturing companies are becoming increasingly aware of integrating circular design strategies (e.g., design-for-disassembly) into product design, adherent guidelines to secure non-destructive disassembly during remanufacturing or refurbishing could be stored and accessed in such a passport, which could not only feature a stepwise walkthrough but also highlight critical awareness points (e.g., when handling hazardous substances). In addition to the condition of components, which is particularly guided by use-phase data and thorough inspection procedures, the technical scope of remanufacturing and refurbishment projects is further reliant on environmental and financial parameters.

To support the decision to select components for a second life, manufacturers see value in using a digital product passport to access data about environmental life cycle impacts (e.g., carbon footprint) on both a product level and a component level. Combined with the financial value of a given component (e.g., through a bill of materials), this would enable manufacturers to target the components with the highest environmental and financial impact. Closely tied to this scoping process, remanufacturing activities - particularly in the case of long-life products - are also challenged by frequent design changes with little attention to the cross-model compatibility of components. Already now, but more so in upcoming product generations where circular principles are likely to become integrated into design, practitioners recommend that a digital product passport should contain information about the compatibility of components. This would add flexibility to the value retention options and partially remedy the challenges of foraging for post-market materials, as imposed by low product return rates. Finally, in cases where components or materials must be sent to suppliers or third-party recycling companies, respectively, a digital product passport is considered suitable for providing information about the actors involved in the production and circulation of a given product, thereby identifying subsequent trajectories in a cascaded flow of products.

Fig. 4 adds further detail to these findings. Aside from product identification measures, priority data include data of high sensitivity, which relates to product and materials (e.g., bill of materials, material composition and recycled content). Additionally, running hours and power, guidelines for non-destructive disassembly and a list of suppliers are considered of high priority. For manufacturers, significant effort is required to generate data and to inform about environmental footprint, availability of customer return channels and waste sorting guidelines, among others. Similarly, additional work is required to

organise a service log. On the other hand, volume and location are considered out of scope, while country of origin is considered to be supportive data.

#### 4.4. Supplier: remanufacture of components

Similar to product manufacturers, supplier engagement in closing resource loops is conditioned by reverse product and material flows, which provide new sourcing opportunities. Either components may be returned as an extension of the manufacturer's take-back programmes, or they are harvested from scrap vendors. In both cases, better access to high-quality data about a component (as facilitated by a digital product passport) is considered a key enabler for higher degrees of value utilisation. For example, rather than remelting - and thus recycling aluminium and steel components, certain types of data may create fruitful conditions for exploring remanufacturing options. As exemplified by interviewees, such data could concern material composition, as well as a list of hazardous substances. Even in closed loops where any given supplier receives its own components, detailed insight into material composition is needed due to frequently occurring design changes and chemical thresholds, as induced by changing regulation. Related to this, suppliers consider a digital product passport a coveted platform for accessing data about recycled content. Although it is not considered essential, this could support suppliers in securing the quality and durability of remanufactured components, as well as tracking such content throughout multiple life cycles.

This is summarised in Fig. 5. While material composition and a list of hazardous substances are considered priority data, data about recycled content is categorised as supportive. Despite being frequently mentioned as a point of awareness by suppliers, the external environment is considered of little importance in this study, which is arguably caused by the robustness of their respective components (e.g., casted aluminium parts). Combined with low availability, external environment is categorised as out of scope in the decision-making context of suppliers.

4.5. Third-party recycling company: retaining material value with reversible losses

Mechanical recycling, as represented in this study, slightly differentiates itself from other activities as the trajectory for products and



Fig. 5. Data to support suppliers in remanufacturing components.

materials is largely determined. However, within this scope, third-party recycling companies find themselves in continuous pursuit of increased quality of recycled materials. Access to certain types of data is expected to support such ambitions. Although pricing mechanisms incentivise manufacturers to sort materials into clean fractions, recycling companies often end up spending a significant amount of time disassembling products and sorting materials themselves before they are shredded. Being able to access disassembly guidelines and material composition in a digital product passport would enable recycling companies to generate even cleaner material fractions. In turn, this would improve the quality of the output, as increased transparency would enable the swift identification of hazardous substances (e.g., from batteries). The lack of data about material composition is a prevalent challenge for recycling companies, particularly in the case of plastics, where various types of plastics, chemical additives and fillers determine their recyclability in terms of reversible or irreversible property losses. Although plastic recycling symbols support this to a certain degree, they are rarely found on plastic components in complex products. To remedy this, data sheets are occasionally attached to the returned materials, yet without a detailed description of their compositions. As argued by practitioners, elaborating on such data sheets and integrating them into a digital product passport would provide coveted data exchange in a systematic manner, thus supporting more effective decision-making in the pursuit of high-quality recycling.

As shown in Fig. 6, data about material composition, a list of hazardous substances and guidelines for non-destructive disassembly are considered priority data, while waste sorting guidelines require additional work and are thus categorised as focused development data.

#### 5. Discussion

Although the needs for data and information in a digital product passport have been scarcely studied in a systematic way, particularly from a multi-actor perspective, a few studies and reports from the academic and the grey literature point towards relevant data points to be included, albeit with only scant empirical foundation. As presented in Table 6, this section aims to juxtapose these suggestions with those presented in this study to highlight similarities and differences, thus both nuancing data points and aiding to materialise the frequently mentioned ones as essential contents of a digital product passport.

Compared to most extant reports and studies, this study identifies several data points, all of which are considered valuable to supplychain actors as levers for the development of decision-making capabilities in a circular economy. This is likely to reflect the industrial embeddedness of the study, which is in contrast to Adisorn et al. (2021) and Berger et al. (2022). By situating the object of enquiry within a small-scale sample of a supply chain, in which actors are becoming increasingly attentive towards the circular transition, practical experiences with a lack of data and information become further augmented. The extant reports and studies depict a dispersed overview of the data and information required for a digital product passport. Frequently mentioned data include a list of hazardous substances, environmental footprint, material composition and product identification measures (Berger et al., 2022; Götz et al., 2022). Based on their perceived importance and partial availability, this study corroborates such findings but nuances them by situating data points in the decision-making contexts of multiple actors. A similar tendency is found in the case of guidelines for non-destructive disassembly, although the option of digitising existing guidelines and manuals has generally received little attention. In the existing studies and reports, data related to usage and maintenance are scantly mentioned, which may indicate a slight incline towards making digital product passports a static platform that captures relevant product data during upstream activities, as argued by Adisorn et al. (2021). However, this study finds that accessing data about running hours and power as well as the service log constitutes essential decision-making support for service providers and manufacturers to unlock inner looping strategies like repair or remanufacture, which may advocate for a more dynamic platform. In terms of a reverse flow of products, neglecting such data may result in a digital product passport that favours third-party recycling companies, as they heavily rely on data from upstream activities, rather than supporting the engagement of a broad array of actors as emphasised by the European Commission (2022) and Berger et al. (2022).



Fig. 6. Data to support third-party recycling companies in retaining material value with reversible losses.

#### Table 6

Comparing identified data points to extant studies and reports.

	Academic literature				
	(Adisorn et	(Berger et	(European	(Götz et	(Danish Business
	al., 2021)	al., 2022)	Commission,	al., 2022)	Authority, 2021)
			2022)		
Usage and maintenance					
Running hours and power		✓			✓
Location				$\checkmark$	
External environment					
Service log		$\checkmark$			$\checkmark$
Product identification					
Serial and product number	√	✓		✓	
Manufacturing date	$\checkmark$	$\checkmark$		$\checkmark$	
Country of origin		$\checkmark$		$\checkmark$	$\checkmark$
Products and materials					
Bill of materials		$\checkmark$			$\checkmark$
Number of life cycles					
Volume					
Material composition	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	
List of hazardous substances	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
Recycled content		$\checkmark$	$\checkmark$		$\checkmark$
Compatibility of					
components across models					
Availability of spare parts			$\checkmark$		
Average lifetime expectancy		$\checkmark$			$\checkmark$
Guidelines and manuals					
Guidelines for non-	√	✓	√	√	
destructive disassembly					
Product manual					
Service manual	$\checkmark$				
Installation guideline					
Safety instructions					
Waste sorting guidelines	$\checkmark$			$\checkmark$	
Supply chain and reverse log	istics				
Suppliers of components		√		✓	
Recycling partners		$\checkmark$			
Customer return channels					
Environmental impact					
Environmental footprint	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
Compliance					
Product-specific regulation	$\checkmark$		$\checkmark$		
(RoHS/REACH)					
Labour conditions		$\checkmark$		$\checkmark$	

Examining data needs further reveals a pattern that the more open the decision-making process is i.e. more options under consideration, the more data does it rely upon. This is clearly illustrated in the juxtaposition of manufacturers and third-party recycling companies. In the case of product take-back, manufacturers inspect returns to make an assessment of the suitable value-retention strategy, either internally or externally, which is contingent on multiple factors, including the external environment and repair history, as also argued by Berger et al. (2022). By contrast, third-party recycling companies experience less uncertainty as the trajectory of materials is pre-determined, particularly in the case of mechanical recycling, as material fractions are shredded and thoroughly sorted afterwards. Consequently, only a few types of data would support recycling companies in conducting high-quality recycling.

#### 5.1. Implications for policymakers

In order to ease industrial adoption, policymakers face the task of striking a balance between setting mandatory requirements for data in a digital product passport – and thus pushing for increased transparency – and protecting intellectual property rights and permitting actors to withhold critical data. This study supports policymakers in doing so through three findings. First, by exploring the data needs from a multi-actor perspective, this study finds that actors have differentiated needs depending on their positions in a

circular supply chain. Highlighting such needs can assist the recently established working groups in defining mandatory data requirements to support decision-making throughout product life cycles. Second, the study confirms that certain types of data, such as material composition, bill of materials or number of life cycles, are considered fairly sensitive. Combining these two aspects calls for multiple levels of accessibility, tailored to the respective needs of each group of actors. This is further argued to increase the attractiveness of a digital product passport, as it removes the risk of data being misused, for example, by competitors. Third, another key consideration for policymakers refers to the rollout of digital product passports. This study contributes to this by examining the availability of data. While certain data (e.g., product identification measures and manuals) are readily available, data related to environmental performance and circularity require additional prioritisation. Therefore, this study corroborates the argument from Götz et al. (2022) that the implementation of digital product passports should be initiated at a small scale - with selected data needs and with flexibility for companies to iteratively develop and refine their data sharing capabilities.

#### 5.2. Implications for industrial managers

From a resource-utilisation perspective, companies could benefit from rethinking data management in favour of a circular supply chain already now, and should thus start preparing themselves on certain aspects – at both a company level and a supply-chain level - for changing institutional settings. What is required, however, differs along the circular supply chain, and this study provides guidance to industrial managers as to how to prioritise their data management throughout product life cycles. Therefore, the findings act as a communicative vessel to enable companies to engage in discussions - within as well as beyond organisational boundaries - about how they can support each other in organising an appropriate data flow. In terms of upstream activities, suppliers and manufacturers are encouraged to assess the availability of relevant data (e.g., on material composition, recycled content or environmental footprint) and explore the possibility of digitising manuals and guidelines. These should be matched with current needs to detect gaps and thus identify starting points for further examination. Furthermore, manufacturers and service providers are encouraged to co-organise new routines of sharing the service log, which are found to benefit both parties. For data related to the use phase (e.g., external environmental conditions), a manufacturer argued that customers are unlikely to be aware of potentially hazardous environmental conditions. For a long-term solution on an industrial scale, manufacturers could consider integrating sensors to detect certain environmental conditions as well as tracking other types of use-phase data, such as running hours and power. Customers should not only request additional data, thus incentivising manufacturers to generate and share it, but preferably engage in updating service logs as well as reversing product flows by exploring end-of-life trajectories, including the availability of product take-back systems.

#### 5.3. Theoretical contribution

Due to the novelty of the policy instrument, industry reports and the academic discourse share the objective of conceptualizing digital product passports in broad terms with little attention to the industrial implications. This study extends the current debate through two central contributions. First, we identify data needs for digital product passports, as suggested by Götz et al. (2022), and situate such needs within five industrial decision-making contexts. Second, we analyse the underlying aspects of importance, availability, and sensitivity for each respective data point to divide them into four groups, namely priority data, focused development data, supportive data, and out-of-scope data. For both contributions, we align with extant studies (e.g. Berger et al., 2022) on exploring data needs for a digital product passport rather than awaiting data *requirements* from regulatory bodies, yet we expand the scope by including perspectives from five groups of circular supply chain actors. By accentuating the differentiating needs for data, we further challenge the current academic discourse on digital product passports, which favours static upstream data for decision-making support over dynamic use-phase data. In terms of value retention, we argue that both are essential features for unlocking a cascaded product flow, thus both supporting the recovery of the functional value as well as the material value.

Furthermore, in a research field that is constantly in movement, the contributions construct a reference point to concretize – and ultimately advance – the discourse on data-driven decisionmaking for a circular economy on two particular areas of research. First, insights into the differentiated data needs are expected to enable further research on system requirements for digital product passports to substantialise, as a concretisation of coveted data qualifies subsequent discussions on interoperability and levels of access. Second, by studying the exchange of data and the utilisation hereof, as facilitated by a digital product passport, from a circular supply chain perspective, the findings offer guidance for improving intra- and interorganizational knowledge management, which are considered essential collaborative and coordinative building blocks towards a systemic transition (Farooque et al., 2019).

#### 6. Conclusion

This study set out to explore the research question: What are the critical decision points and data needs for actors in a circular supply chain to support product life cycle decision-making?

In the pursuit of this, five decision-making contexts have been identified, each of which represents a core activity from an essential actor in a circular supply chain. In all cases, enhanced transparency and better access to high-quality data, as facilitated by a digital product passport, is considered valuable for product life cycle decision-making. More specifically, a total of 28 data points have been identified, clustered into seven categories: (1) usage and maintenance, (2) product identification, (3) product and materials, (4) guidelines and manuals, (5) supply chain and reverse logistics, (6) environmental data and (7) compliance. Linking these to the five decision-making contexts illustrates how data needs for utilising digital product passports differ along the circular supply chain. Subsequently, the data points are assessed in terms of their perceived importance, availability and sensitivity. This enables a categorisation of data points to support practitioners in strengthening their data management, while providing policymakers with industrial perspectives on the contents of digital product passports. Both are further validated by comparing the identified data points with extant studies and reports.

Reflecting on the scope of this study, certain limitations must be highlighted. Concerning the objective of exploring data needs for a circular supply chain, authorities were not included despite their important role in conducting market surveillance and securing coherence across policy instruments. Likewise, the presented data needs support circularity at the actor level; they do not monitor or improve sustainability at a higher level. Furthermore, this study is situated in the mechatronics industry. Data needs are likely to differ across industries, albeit with a share of similarities, which brings their generalisability into question.

As the scientific debate is only starting to take shape, several avenues for further research can be outlined. First, it is relevant to explore cross-industrial similarities and differences in terms of data needs. This would both elaborate on the findings from this study and materialise the contents of digital product passports. Second, it is yet to be assessed how the availability of digital product passports can support decision-making as part of a circular economy. Third, it is considered relevant to understand the driving mechanisms and potential challenges for industries to adopt digital product passports. Fourth, further research is suggested to explore system requirements by utilising the findings from this study to illuminate the areas of interoperability and data governance and the facilitation of multiple levels of access, including through the lens of blockchain technology.

#### **Declaration of competing interest**

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

#### Acknowledgements

This research is funded by the Manufacturing Academy of Denmark (MADE) as part of the innovation programme MADE Fast (Funding nr.: 9090-00002B). The authors wish to thank all the companies and interviewees involved for their participation throughout this study.

#### Sustainable Production and Consumption 37 (2023) 242-255

#### Appendix A. Interview guide for data collection

Theme	Question
Introduction	• What is your role at your company?
Digital product passports	<ul> <li>When hearing about this digital product passport, what are your minimulate thoughts?</li> <li>Do you believe that better access to data and information could strengthen the circular transition?</li> <li>When the pass 2</li> </ul>
	• In which situations do you believe that a digital product passport can support you?
Creating a scenario for a decision-making context that fits the ro	le of the interviewee.
The first question in the following is an example of a supplier:	
Needs for data and information	• Let us imagine that a product manufacturer receives a worn-out product from the market, disassembles it
	and sends a component to you to make it function again. Which types of data could support you to do so?
	Who should provide this information?
	• To what extent does this already exist?
	<ul> <li>Are some types of data more important than other types of data?</li> </ul>
Systems	<ul> <li>How important is it that a digital product passport can communicate with other systems?</li> <li>O Why?</li> </ul>
	<ul> <li>Which systems should a digital product passport communicate with?</li> </ul>
Outlook	• Do you feel that you are ready to explore and implement a digital product passport?
	• Do you see it as a valuable tool or an administrative burden?
	• Which drivers do you see towards implementing digital product passports?
	Which barriers do you see to implementing digital product passports?

#### References

- Adisorn, T., Tholen, L., Götz, T., 2021. Towards a digital product passport fit for contributing to a circular economy. Energies 14. https://doi.org/10.3390/en14082289.
- Ahmad, R., Kamaruddin, S., 2012. An overview of time-based and condition-based maintenance in industrial application. Comput. Ind. Eng. 63, 135–149. https://doi.org/10. 1016/j.cie.2012.02.002.
- Antikainen, M., Uusitalo, T., Kivikytö-Reponen, P., 2018. Digitalisation as an enabler of circular economy. Procedia CIRP 73, 45–49. https://doi.org/10.1016/j.procir.2018.04. 027.
- Awan, U., Shamim, S., Khan, Z., Zia, N.U., Shariq, S.M., Khan, M.N., 2021. Big data analytics capability and decision-making: the role of data-driven insight on circular economy performance. Technol. Forecast. Soc. Chang. 168, 120766. https://doi.org/10.1016/j. techfore.2021.120766.
- Ayati, M.S., Shekarian, E., Majava, J., Wæhrens, B.V., 2022. Toward a circular supply chain: understanding barriers from the perspective of recovery approaches. J. Clean. Prod. 359, 1–18. https://doi.org/10.1016/j.jclepro.2022.131775 131775.
- Baxter, P., Jack, S., 2008. Qualitative case study methodology: study design and implementation for novice researchers. TQR https://doi.org/10.46743/2160-3715/2008.1573.
- Berger, K., Schöggl, J.-P., Baumgartner, R.J., 2022. Digital battery passports to enable circular and sustainable value chains: conceptualization and use cases. J. Clean. Prod. 353, 131492. https://doi.org/10.1016/j.jclepro.2022.131492.
- Bockholt, M.T., Hemdrup Kristensen, J., Colli, M., Meulengracht Jensen, P., Vejrum Wæhrens, B., 2020. Exploring factors affecting the financial performance of end-oflife take-back program in a discrete manufacturing context. J. Clean. Prod. 258, 120916. https://doi.org/10.1016/j.jclepro.2020.120916.
- Bressanelli, G., Adrodegari, F., Perona, M., Saccani, N., 2018. The role of digital technologies to overcome circular economy challenges in PSS business models: an exploratory case study. Procedia CIRP 73, 216–221. https://doi.org/10.1016/j.procir.2018.03.322.
- Calicchio Berardi, P., Peregrino de Brito, R., 2021. Supply chain collaboration for a circular economy from transition to continuous improvement. J. Clean. Prod. 328, 129511. https://doi.org/10.1016/j.jclepro.2021.129511.
- Chauhan, C., Parida, V., Dhir, A., 2022. Linking circular economy and digitalisation technologies: a systematic literature review of past achievements and future promises. Technol. Forecast. Soc. Chang. 177, 121508. https://doi.org/10.1016/j.techfore.2022. 121508.
- Chen, L., Zhao, X., Tang, O., Price, L., Zhang, S., Zhu, W., 2017. Supply chain collaboration for sustainability: a literature review and future research agenda. Int. J. Prod. Econ. 194, 73–87. https://doi.org/10.1016/j.ijpe.2017.04.005.
- Danish Business Authority, 2021. Looping on Data Best Practices and Barriers for Sharing Data in Circular Business Models.
- Eisenhardt, K.M., Graebner, M.E., 2007. Theory building from cases: opportunities and challenges. AMJ 50, 25–32. https://doi.org/10.5465/amj.2007.24160888.
- Ellen MacArthur Foundation, 2012. Towards the Circular Economy: Economic and Business Rationale for an Accelerated Transition.
- European Commission, 2022. Communication From the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of The Regions – On Making Sustainable Products the Norm.
- European Commission, 2019. The European Green Deal.
- Farooque, M., Zhang, A., Thürer, M., Qu, T., Huisingh, D., 2019. Circular supply chain management: a definition and structured literature review. J. Clean. Prod. 228, 882–900. https://doi.org/10.1016/j.jclepro.2019.04.303.
- Flyvbjerg, B., 2006. Five misunderstandings about case-study research. Qual. Inq. 12, 219–245. https://doi.org/10.1177/1077800405284363.

- Geissdoerfer, M., Savaget, P., Bocken, N.M.P., Hultink, E.J., 2017. The circular economy a new sustainability paradigm? J. Clean. Prod. 143, 757–768. https://doi.org/10.1016/j. jclepro.2016.12.048.
- Ghisellini, P., Cialani, C., Ulgiati, S., 2016. A review on circular economy: the expected transition to a balanced interplay of environmental and economic systems. J. Clean. Prod. 114, 11–32. https://doi.org/10.1016/j.jclepro.2015.09.007.
- Giovanardi, M., Konstantinou, T., Pollo, R., Klein, T., 2023. Internet of things for building facade traceability: a theoretical framework to enable circular economy through lifecycle information flows. J. Clean. Prod. 382, 135261. https://doi.org/10.1016/j. iclepro.2022.135261.
- González-Sánchez, R., Settembre-Blundo, D., Ferrari, A.M., García-Muiña, F.E., 2020. Main dimensions in the building of the circular supply chain: a literature review. Sustainability 12. https://doi.org/10.3390/su12062459.
- Götz, T., Berg, H., Jansen, M., Adisorn, T., Cembrero, D., Markkanen, S., Chowdhury, T., 2022. Digital Product Passport: The Ticket to Achieving a Climate Neutral and Circular European Economy?
- Goyal, S., Esposito, M., Kapoor, A., 2018. Circular economy business models in developing economies: lessons from India on reduce, recycle, and reuse paradigms. Thunderbird Int. Bus. Rev. 60, 729–740. https://doi.org/10.1002/tie.21883.
- Graedel, T.E., Allwood, J., Birat, J.-P., Buchert, M., Hagelüken, C., Reck, B.K., Sibley, S.F., Sonnemann, G., 2011. What do we know about metal recycling Rates? J. Ind. Ecol. 15, 355–366. https://doi.org/10.1111/j.1530-9290.2011.00342.x.
- Hopkinson, P., Zils, M., Hawkins, P., Roper, S., 2018. Managing a complex global circular economy business model: opportunities and challenges. Calif. Manag. Rev. 60, 71–94. https://doi.org/10.1177/0008125618764692.
- Hsieh, H.-F., Shannon, S.E., 2005. Three approaches to qualitative content analysis. Qual. Health Res. 15, 1277–1288. https://doi.org/10.1177/1049732305276687.
- Huang, Y., Wang, Z., 2017. Values of information sharing: a comparison of supplierremanufacturing and manufacturer-remanufacturing scenarios. Transport Res E-Log 106, 20–44. https://doi.org/10.1016/j.tre.2017.07.015.
- Ingemarsdotter, E., Jamsin, E., Balkenende, R., 2020. Opportunities and challenges in IoTenabled circular business model implementation – a case study. Resour. Conserv. Recycl. 162, 105047. https://doi.org/10.1016/j.resconrec.2020.105047.
- Jäger-Roschko, M., Petersen, M., 2022. Advancing the circular economy through information sharing: a systematic literature review. J. Clean. Prod., 133210. https://doi.org/10. 1016/j.jclepro.2022.133210.
- Jensen, S.F., Kristensen, J.H., Uhrenholt, J.N., Rincón, M.C., Adamsen, S., Waehrens, B.V., 2022. Unlocking barriers to circular economy: an ISM-based approach to contextualizing dependencies. Sustainability 14. https://doi.org/10.3390/su14159523.
- Kirchherr, J., Reike, D., Hekkert, M., 2017. Conceptualizing the circular economy: an analysis of 114 definitions. Resour. Conserv. Recycl. 127, 221–232. https://doi.org/10. 1016/j.resconrec.2017.09.005.
- Kristoffersen, E., Blomsma, F., Mikalef, P., Li, J., 2020. The smart circular economy: a digitalenabled circular strategies framework for manufacturing companies. J. Bus. Res. 120, 241–261. https://doi.org/10.1016/j.jbusres.2020.07.044.
- Leising, E., Quist, J., Bocken, N., 2018. Circular economy in the building sector: three cases and a collaboration tool. J. Clean. Prod. 176, 976–989. https://doi.org/10.1016/j. jclepro.2017.12.010.
- Lopes de Sousa Jabbour, A.B., Rojas Luiz, J.V., Rojas Luiz, O., Jabbour, C.J.C., Ndubisi, N.O., Caldeira de Oliveira, J.H., Junior, F.H., 2019. Circular economy business models and operations management. J. Clean. Prod. 235, 1525–1539. https://doi.org/10.1016/j. jclepro.2019.06.349.
- Lüdeke-Freund, F., Gold, S., Bocken, N.M.P., 2019. A review and typology of circular economy business model patterns. J. Ind. Ecol. 23, 36–61. https://doi.org/10.1111/jiec. 12763

- MacDonald, E.F., She, J., 2015. Seven cognitive concepts for successful eco-design. J. Clean. Prod. 92, 23–36. https://doi.org/10.1016/j.jclepro.2014.12.096.
- Ondemir, O., Gupta, S.M., 2014. Quality management in product recovery using the internet of things: an optimization approach. Comput. Ind. 65, 491–504. https://doi.org/ 10.1016/j.compind.2013.11.006.
- Pagoropoulos, A., Pigosso, D.C.A., McAloone, T.C., 2017. The emergent role of digital Technologies in the Circular Economy: a review. Procedia CIRP 64, 19–24. https://doi.org/ 10.1016/j.procir.2017.02.047.
- Pedersen, E., Remmen, A., 2022. Challenges with product environmental footprint: a systematic review. Int. J. Life Cycle Assess. 27, 342–352. https://doi.org/10.1007/s11367-022-02022-3.
- Plociennik, C., Pourjafarian, M., Nazeri, A., Windholz, W., Knetsch, S., Rickert, J., Ciroth, A., Precci Lopes, A.do C., Hagedorn, T., Vogelgesang, M., Benner, W., Gassmann, A., Bergweiler, S., Ruskowski, M., Schebek, L., Weidenkaff, A., 2022. Towards a digital lifecycle passport for the circular economy. Procedia CIRP 105, 122–127. https://doi. org/10.1016/j.procir.2022.02.021.

Preston, F., 2012. A Global Redesign? Shaping the Circular Economy. 2012, p. 22.

- Rejeb, A., Suhaiza, Z., Rejeb, K., Seuring, S., Treiblmaier, H., 2022. The internet of things and the circular economy: a systematic literature review and research agenda. J. Clean. Prod. 350, 131439. https://doi.org/10.1016/j.jclepro.2022.131439.
- Rosa, P., Sassanelli, C., Urbinati, A., Chiaroni, D., Terzi, S., 2020. Assessing relations between circular economy and industry 4.0: a systematic literature review. Int. J. Prod. Res. 58, 1662–1687. https://doi.org/10.1080/00207543.2019.1680896.
- Serna-Guerrero, R., Ikonen, S., Kallela, O., Hakanen, E., 2022. Overcoming data gaps for an efficient circular economy: a case study on the battery materials ecosystem. J. Clean. Prod. 374, 133984. https://doi.org/10.1016/j.jclepro.2022.133984.

- Stahel, W.R., 2016. The circular economy. Nature 531, 435–438. https://doi.org/10.1038/ 531435a.
- Subramanian, R., Gupta, S., Talbot, B., 2009. Product design and supply chain coordination under extended producer responsibility. Prod. Oper. Manag. 18, 259–277.
- Sudusinghe, J.I., Seuring, S., 2022. Supply chain collaboration and sustainability performance in circular economy: a systematic literature review. Int. J. Prod. Econ. 245, 108402. https://doi.org/10.1016/j.ijpe.2021.108402.
- Taherdoost, H., 2016. Sampling methods in research methodology; how to choose a sampling technique for research. SSRN J. https://doi.org/10.2139/ssrn.3205035.
- Takata, S., Kirnura, F., van Houten, F.J.A.M., Westkamper, E., Shpitalni, M., Ceglarek, D., Lee, J., 2004. Maintenance: changing role in life cycle management. CIRP Ann. 53, 643–655. https://doi.org/10.1016/S0007-8506(07)60033-X.
- Tseng, M.-L., Chiu, A.S.F., 2013. Evaluating firm's green supply chain management in linguistic preferences. J. Clean. Prod. 40, 22–31. https://doi.org/10.1016/j.jclepro.2010. 08.007.
- Uhrenholt, J.N., Hemdrup, J.H., Adamsen, S., Jensen, S.F., Colli, M., Waehrens, B.V., 2022. Twin transition: synergies between circular economy and internet of things - a study of Danish manufacturers. Circ. Econ. 1.
- Walden, J., Steinbrecher, A., Marinkovic, M., 2021. Digital product passports as enabler of the circular economy. Chem. Ing. Tech. 93, 1717–1727. https://doi.org/10.1002/cite. 202100121.
- Xiong, Y., Zhao, Q., Zhou, Y., 2016. Manufacturer-remanufacturing vs supplierremanufacturing in a closed-loop supply chain. Int. J. Prod. Econ. 176, 21–28. https://doi.org/10.1016/j.ijpe.2016.03.001.
- Yin, R.K., 2009. Case Study Research: Design and Methods. SAGE.